Check for updates

OPEN ACCESS

EDITED BY Mark Altaweel, University College London, United Kingdom

REVIEWED BY Hari Harjanto Setiawan, National Research and Innovation Agency (BRIN), Indonesia Mayssoun Sukarieh, King's College London, United Kingdom

*CORRESPONDENCE Natalia Triana Ángel ⊠ n.triana@cgiar.org

RECEIVED 27 May 2024 ACCEPTED 19 July 2024 PUBLISHED 06 August 2024

CITATION

Díaz Baca MF, Moreno Lerma L, Burkart S and Triana Ángel N (2024) Why do rural youth migrate? Evidence from Colombia and Guatemala. *Front. Sociol.* 9:1439256. doi: 10.3389/fsoc.2024.1439256

COPYRIGHT

© 2024 Diaz Baca, Moreno Lerma, Burkart and Triana Ángel. This is an open-access article distributed under the terms of the Creative Commons Attribution License (CC BY). The use, distribution or reproduction in other forums is permitted, provided the original author(s) and the copyright owner(s) are credited and that the original publication in this journal is cited, in accordance with accepted academic practice. No use, distribution or reproduction is permitted which does not comply with these terms.

Why do rural youth migrate? Evidence from Colombia and Guatemala

Manuel Francisco Díaz Baca^{1,2}, Leonardo Moreno Lerma³, Stefan Burkart² and Natalia Triana Ángel²*

¹Department of International Agricultural Policy and Environmental Governance, University of Kassel, Kassel, Germany, ²Tropical Forages Program, International Center for Tropical Agriculture (CIAT), Cali, Colombia, ³Independent Researcher, Cali, Colombia

Migration, from rural to urban settings is a common phenomenon in Latin America, due to social, economic, political, and other factors. Young people in search of economic and educational opportunities, financial, and social stability, have been migrating to larger urban centers, thus crafting important shifts in rural labor, generational transfer, and domestic economies. Through a systematic literature review of scientific literature, and documents from public institutions and international organizations, published between 2012 and 2022, this article addresses rural-urban migration of youth in Colombia and Guatemala's cattle sector, particularly identifying (i) driving factors, (ii) their impacts on cattle farming, and (iii) public policies implemented to counteract prejudicial effects. Results show that unemployment, lack of educational opportunities, and insecurity are the main reasons for youth migration to cities or abroad, with Mexico, the United States, and Spain being the most common destinations. Additionally, impacts on the cattle sector include shortage of labor and a perfectible generational transfer, hindering the modernization of the industry and investments in climate change adaptation and mitigation strategies. Despite various implemented public policies, the results are partial, and the issue of accelerated youth migration remains relevant. Consequently, without more effective measures adopted by national governments, the cattle sector will lag behind its regional and international competitors, deterring the achievement of the Sustainable Development Goals. As the main contribution of the study, the analysis of migration is highlighted based on its effects on a specific economic sector and not focused on its causes, as evidenced in a wide range of literature.

KEYWORDS

rural migration, public policy, climate change mitigation, climate change adaptation, sustainability, competitiveness

Highlights

- The origins of migration in Colombia and Guatemala are concentrated in rural areas.
- Limited labor availability in the cattle sector is aggravated by youth migration.
- Policies identify the drivers of migration and define inclusion criteria for youth.

1 Introduction

Human migration has been a constant from ancient to modern times (Sirbu et al., 2021), driven by macro-social, political, economic, and environmental factors (Castelli, 2018). Latin America is not exempt from this phenomenon, currently serving as both an origin and transit region for migrants (International Organization for Migration, 2020). In 2018, 1.3 million people from Latin America and the Caribbean migrated to countries in the Organization for Economic Cooperation and Development (OECD), with the United States and Spain being their main destinations (Inter-American Development Bank, 2021). Additionally, there is an intraregional migration process, as four out of every five migrants from South America reside in another country within the subregion (International Organization for Migration, 2022a).

Among the distinct types of migration, the movement from rural to urban areas stands out, particularly among young people seeking to escape precarious socio-economic conditions (Zabala, 2021). In Latin America, out of the 31 million rural youth aged 15-29, 11.9 million are unemployed, experiencing higher poverty levels compared to adults in those regions and individuals of the same age in urban areas (OECD and FAO, 2019). Academic literature indicates that this age group is more likely to migrate compared to adults, with males surpassing females in this regard (Prieto et al., 2022). While youth immigration has positive repercussions in the destination, such as rejuvenating the population, it has adverse effects on the place of origin by accelerating aging and, consequently, reducing the working-age population (García et al., 2019). One of the most affected sectors is agriculture, facing challenges like labor shortages (Rosendo et al., 2019), hampering the productivity of rural activities due to the difficulty for older farmers to adopt and finance the latest technologies (Zabala, 2021).

This research aims to address the phenomenon of youth migration and its impacts on the agricultural sector in Colombia and Guatemala. It is noteworthy that the focus is on the cattle sector due to its significance for Latin America and the Caribbean, contributing over 14% of the total agricultural production value (FAO, 2018), serving as a crucial source of income for the rural population, and generating employment for millions of people. Cattle farming is also vital for the region, aiding in achieving Sustainable Development Goals (SDGs), i.e., SDG-1 (no poverty), SDG-2 (zero hunger), SDG-6 (gender equality), SDG-8 (decent work and economic growth), SDG-10 (reduced inequalities), SDG-13 (climate action), among others (United Nations, 2018). Consequently, establishing connections between youth migration and cattle farming not only helps understand the issues affecting this population but also the negative impacts migration is having on a fundamental sector in economic, environmental, and social terms.

The selection of these countries aligns with several criteria: (i) Both rank high among countries with the most migrants in their regions: Colombia is second in South America, surpassed only by Venezuela, and Guatemala is second in Central America after El Salvador (Statista, 2023a); (ii) Colombia designed the Sustainable Cattle Policy, while Guatemala implemented the National Strategy for Sustainable Low-Emission Cattle Farming, making it relevant to analyze whether these policies have considered the phenomenon of youth migration (Moreno et al., 2022; Díaz et al., 2024); and (iii) Guatemala and Colombia are the second and third countries receiving the most remittances as a result of migration, with US\$18.11 billion and US\$9.44 billion annually, respectively, only surpassed by Mexico with US\$60.3 billion (Statista, 2023b).

The study aims to address two gaps in research, (i) although there is abundant literature on youth migration in some of the region's countries, such as Mexico (Velasco, 2016; Suárez, 2023) and Brazil (Gaspar and Chatti, 2022; Spanevello et al., 2022), others have yet received less attention, for which the present study prioritizes two of the less-covered countries; and (ii) the migration topic has been primarily studied from its driving factors (Castelli, 2018; Barrios et al., 2022), gender perspectives (Everaert, 2021), or focus on population groups, such as youth (Cazzuffi and Fernández, 2018) (issues also addressed in this analysis), but is seldomly related to its effects on the cattle sector, which not only presents itself as a novelty in research but also proposes thematic lines for future investigations. Thus, this article addresses the following research question: what are the socioeconomic drivers and impacts on the cattle sector of youth migration in Colombia and Guatemala, and how is the issue countered, if so, through the implementation of public policies?

The study is divided into the following sections: section 2 introduces fundamental concepts such as youth, migration, rurality, public policies, and laws, along with diverse theories of migration and the kaleidoscope model of policy change. Section 3 outlines the study's characteristics, i.e., focus, stages, sources of information, and limitations. Section 4 identifies and analyzes the main drivers of migration, its effects, and implemented public policies. Section 5 provides methodological and practical recommendations and Section 6 formulates general conclusions.

2 Theoretical framework

To develop the stated objectives, it is necessary to clarify some concepts, such as *youth*, *rurality*, *laws*, and *public policies*. Additionally, reference is made to four theories of migration and the *Kaleidoscope Model of Policy Change*.

Youth refers to a diverse age range which is defined based on differing perspectives. They can be organized into two main groups, namely the national perspective, which is dependent on each country's parameters, and the perspective of international organizations (Sandoval et al., 2022). In Colombia, a young person is anyone between 14 and 28 years of age, who is undergoing physical, intellectual, social, economic, and cultural development (Congreso de la República de Colombia, 2018). In Guatemala, the range is broader, involving individuals between 13 and 30 years old, divided into four groups: 13–17 years (adolescence), 18–20 years (late adolescence), 20–25 years (fully young adults), and 26–30 years (young adults) (Consejo Nacional de la Juventud y Fondo de Población de las Naciones Unidas, 2020). United Nations (2014) limit this concept to individuals between 15 and 24 years of age.

The term *rurality* often alludes to notions of the countryside, agriculture, and isolated, sparsely populated areas in contrast to the city, industry, and population density (Morales et al., 2015). The concept becomes more complex as current academic literature refers to a new rurality where the dichotomy is not as clear, and socio-economic dynamics possess characteristics of both scenarios. In this sense, the concept also encompasses spaces where non-agricultural

activities and rural-urban relationships are strengthened (Chacón, 2021).

It is also important to distinguish between *laws* and *public policies*. While both aim to address various social issues, *laws* emerge from the congresses or parliaments of each country through specific constitutional procedures (Villadangos, 2022), while *public policies* are driven by national or local public institutions, either independently or in collaboration with non-governmental actors (private entities, NGOs, social groups, etc.) (Howlett and Cashore, 2014).

Migration, on the other hand, is defined as the movement of an individual from their place of residence to another, implying a substantial distance and resulting in permanent residence (Toney and Bailey, 2014). *Migration* can be within a region of the same country (internal) or beyond national borders (external), in which case it is termed immigration or emigration (Toney and Bailey, 2014).

On the other hand, the migration phenomenon can be theoretically explained by two variables: push, understood as factors that "push" a person to leave their place of residence, and pull, which are factors that attract, and both can be economic, social, political, environmental, and cultural, among others (Schiavon, 2023). The push-pull theory originated from the work of the German geographer Ernst Georg Ravenstein and was dominant in the first half of the 20th century, incorporating principles of political economy of the time, such as economic rationalism, individualism, and liberalism (García, 2018). The theory remains relevant in studies explaining migration from various perspectives, from migration between cities in China (Fangqu, 2022) to return migration in literature reviews (Mohamed and Abdul, 2020). However, the theory has also been criticized because it tends to list a series of factors that, while fostering migration, do not provide a structural and social explanation of the phenomenon (Hass, 2021), making it necessary to relate it to other theoretical postulates.

It is worth highlighting the new economics of labor migration, which states that (i) the decision to migrate is not individual but rather a risk management strategy of households and families in response to labor market failures, and (ii) there is the possibility that the migrant will return to their place of origin after achieving their goals of savings, investment, capital acquisition, and more (Gheasi and Nijkamp, 2017). It is also essential to refer to the world-systems theory, according to which migration is not a personal decision but the consequence of a structural dependency of poor countries on rich countries, as the latter have achieved higher levels of economic development and have the capacity and need to integrate labor (Morawska, 2021). Finally, the migration systems theory suggests that migration changes both micro factors (personal relationships) and macro factors (economic and social conditions) in the places of origin and destination, understanding the phenomenon in positive terms and in relation to development (Gheasi and Nijkamp, 2017).

To understand how governments have responded to the phenomenon of migration, particularly in the cattle sector, the *Kaleidoscope Model of Policy Change (KMPC)* can be applied. This model proposes a set of stages for analysis, namely (i) agendasetting: determines topics of interest for governments; (ii) design: proposes solutions to the issue; (iii) adoption: considers adverse factors that may hinder policy implementation; (iv) implementation: executes the proposed solutions; and (v) evaluation and reform: establishes whether objectives were achieved or changes are necessary (Resnick et al., 2018). It is worth noting that this theoretical model has been applied in the analysis of nutrition policies in Africa (Hendriks et al., 2017) and fertilizer subsidies in Tanzania (Mather and Ndyetabula, 2016), demonstrating its adaptability to agricultural sector issues.

Considering these elements, it is necessary to make some clarifications. Firstly, in this study, *youth* include individuals aged between 13 and 30 years, allowing for the incorporation of both national and international definitions and encompassing a broader range of research on migration. Similarly, the definition of *new rurality* is invoked since in both Colombia and Guatemala, scenarios are identified that, while not corresponding to the traditional concept of rurality, involve phenomena of youth migration. It is noteworthy that the study simultaneously addresses *laws* and *public policies*, enabling a comprehensive analysis of legislative and governmental actions, while also referring to both internal and external migration.

Regarding the theories, although none directly address the migration-cattle relationship, their different postulates and interrelations construct a sufficiently broad analytical framework to understand the issue. For its part, the KMPC allows investigating whether public policies have responded to the migration phenomenon by counteracting its causes, creating better social conditions that encourage return, overcoming dependency on developed countries, and enhancing positive impacts (Figure 1).

3 Materials and methods

Based on the concept of systematic literature review (Palmatier et al., 2018), this study employed a critical analysis of documentary material regarding the phenomenon of youth migration in Colombia and Guatemala. For this purpose, a qualitative approach was applied, where concepts, characteristics, and descriptions were prioritized (Lune and Berg, 2017). The research followed a set of seven stages, namely (i) idea; (ii) formulation; (iii) definition of the study type; (iv) definition of information sources; (v) data collection; (vi) data analysis and discussion; and (vii) preparation of the final report (Sampieri et al., 2014). Secondary information sources were used (Sahu, 2013), divided into three types: (i) publications from state agencies, particularly public policies; (ii) documents from international organizations, such as FAO, OECD, USDA, providing accurate information on the addressed contexts; and (iii) academic articles to develop sections of the theoretical framework, results, analysis, and discussion. Data collection was carried out using the Google Scholar search engine, considering keywords such as rural migration, youth migration, public policies, and cattle sector. Two inclusion criteria were established, namely (i) publications from 2012 to 2022, to present the most up-to-date bibliography; and (ii) academic articles from indexed journals (Table 1).

Among the study's limitations, four points are noteworthy. First, there was a scarcity of literature addressing the issue of youth migration related to the cattle sector, leading to a reliance on the authors' interpretations to a significant extent. Second, there are a vast number of public policies that directly and indirectly impact the addressed phenomenon, requiring the analysis to focus on national policies and exclude those at the local or regional levels. Third, due to the large number of public policies found, it was not possible to delve deeply into each one, limiting the analysis to a balance of results in each country. Fourth, since the study was limited to Colombia and



Guatemala, generalizing the conclusions to the Latin American level is not possible.

4 Results, analysis, and discussion

4.1 Push-pull factors of migration

As discussed, migration involves push and pull factors and can occur internally or externally. The following factors are presented in the addressed scenarios, particularly concerning rural youth (aged 13–30 years).

In a study conducted in Arauca, Colombia, lack of employment was identified by Mina and Téllez (2022) as the main cause of youth migration, and youth lack sufficient support from public or private actors to start a business or formally enter the job market. The same authors indicate minimal technical-professional training for this population, coupled with excessive requirements for job access. A similar issue was identified by López et al. (2017) for Chinavita, Boyacá, where young migrants decide to leave primarily due to a lack of formal employment opportunities. In Guatemala, the search for better job opportunities is linked to variables such as the high cost of satisfying the basic family needs and widespread poverty (Santibáñez et al., 2017). Factors like low income, limited ability to acquire goods and services, unstable employment, and the responsibility of young people to financially support their parents are also evident (Santibáñez et al., 2017; Spohn, 2017).

The situation is particularly complex for women. In Guatemala, the main reasons for women not working in their place of origin are linked to a strong focus on unpaid household labor (79.1%) or receiving low salaries (16.5%) (Barrios et al., 2022). For girls and young women, internal migration from rural areas to urban centers is common, where they work as domestic workers, often receiving low, but some, payment (Santibáñez et al., 2017). In Colombia, rural

women face a higher burden of unrecognized work, hindering their income generation (Consejo Nacional de Política Económica y Social, 2021). The lack of educational opportunities is also a significant factor, as seen in Guatemalan families migrating to the United States to provide better education for their young members (Spohn, 2017). A similar situation is observed in Colombia, where youth migration spans over generations, with parents stating they migrated as adolescents, just as their children do today (López et al., 2017).

In Colombia, another significant driver for migration is the internal armed conflict, a phenomenon that has affected the country since the 1950s, resulting in human rights violations (Mina and Téllez, 2022). The conflict has particularly affected young people between 18 and 28 years of age, with a national total of 2,172,373 victims accounted for over these decades, including victimizing events such as forced displacement, enforced disappearance, threats, kidnapping, land dispossession, among others (Unidad de Víctimas, 2017). While violence is not the primary factor for youth migration, it is mentioned by both women (10.4%) and men (3.1%) in Guatemala and the Northern Triangle (including El Salvador and Honduras) due to issues like extortion and, in the case of women, domestic violence (Everaert, 2021). In rural areas, the phenomenon of land appropriation by drug traffickers is strong, especially in border areas with little or no military or state presence, and farmers are offered an excessively low price for their land and are then forced to move (Jiménez, 2016). Land disputes also occur between farmers and landowners, with the latter taking advantage of the lack of property records or in alliance with state forces (Aguirre, 2014).

Regarding the regions with the highest migration rates in Colombia, these include Bogotá (26.60%), Antioquia (7.30%), Cundinamarca (5.70%), and Valle del Cauca (5.60%), which are also the major recipients of migrants, with percentages of 14.90, 7.30, 17.20, and 6.40%, respectively (International Organization for Migration, 2021). These metropolitan areas represent pull factors for rural youth, such as salaries that allow them to increase their

TABLE 1 Sources of information.

Article section	Keywords	Country	Public entities	International organizations and statistical databases	Scientific articles
4.1	Migration, youth, work, education, violence	Colombia	 Consejo Nacional de Política Económica y Social (CONPES) Departamento Administrativo Nacional de Estadística de Colombia (DANE) Unidad para las Víctimas 	International Organization for Migration (IOM)	 López et al. (2017) Mina and Téllez (2022) Torres (2021)
		Guatemala	-	 Macrodata International Organization for Migration (IOM) 	 Santibáñez et al. (2017) Spohn (2017) Barrios et al. (2022) Everaert (2021) Jiménez (2016) Aguirre (2014) Lozano et al. (2015) Canales et al. (2016) Cazzuffi and Fernández (2018) González et al. (2022)
4.2	Migration, youth, cattle farming, labor, aging, modernization	Colombia Guatemala	DANE Ministerio de Agricultura y Desarrollo Rural (MADR) Gobierno de la República Ministerio de Agricultura, Pesca y Alimentación (MAPA) Consejo Agropecuario Centroamericano (CAC) Instituto Nacional de Estadística Guatemala (INE)	 Grupo Interagencial sobre Flujos Migratorios Mixtos (GIFMM) Economic Commission for Latin America and the Caribbean (ECLAC) 	 Martínez et al. (2022) Aguilar and Serrano (2015) Triana and Burkart (2023) Villeda (2020) Weller (2016)
		Analysis	-	• ECLAC	Gheasi and Nijkamp (2017)Groher et al. (2020)Morawska (2021)

Díaz Baca et al.

(Continued)

Article section	Keywords	Country	Public entities	International organizations and statistical databases	Scientific articles
6.4	Public policy, youth, Colombia migration, unemployment, education, security	Colombia	 CONPES Congreso de la República MADR Proyectos Integrales de Desarrollo Agropecuario y Rural (PIDAR) Agencia de Desarrollo Rural (ADR) Agencia de Desarrollo Rural (ADR) Ministerio de Educación Nacional (MEN) Servicio Nacional de Aprendizaje (SENA) 	 Food and Agriculture Organization of the United Nations (FAO) 	 Triana and Burkart (2023) Triana et al. (2020)
		Guatemala	 Presidencia de la República Ministerio de Agricultura, Ganadería y Alimentación (MAGA) Ministerio de Trabajo y Prevision Social (MTPS) Consejo Nacional de Seguridad Consejo Nacional de la Juventud 	 United Nations Population Fund CEPAL Programa De Las Naciones Unidas Para El Desarrollo (PNUD) 	1
		Analysis	:	1	García et al. (2019)Valenciano et al. (2016)

consumption and savings, as well as providing economic assistance to their relatives in their places of origin (Torres, 2021). They also concentrate the country's main universities and offer protection from internal armed conflict dynamics (Mina and Téllez, 2022). External migration destinations include the United States (22%), Ecuador (20%), and Spain (14%) (DANE, 2022a; Figure 2). Although most migrants are adults (over 27 years old), young adults (19–26) and teenagers (12–18) represent 17.70 and 2.07% of them (International Organization for Migration, 2021). These destinations serve as pull factors for remittances, as the United States and Spain rank high in the origin of transfers to Colombia, especially to the departments of Valle del Cauca, Cundinamarca, and Antioquia (International Organization for Migration, 2021).

In Guatemala, internal migrants mainly head north in search of farmland or to Guatemala City, where more job opportunities exist (Lozano et al., 2015). Among the municipalities of origin are Santa Lucía La Reforma, Cabricán, and Zacualpa, where 75.86, 30.30, and 26.32% of migrants choose these destinations (Rocha and Gramajo, 2017). External migration happens mostly to the United States (89.65%), Mexico (3.38%), and Belize (1.97%) (Datos Macro, 2020). Notably, Guatemalan migrants are the youngest among the countries of the Northern Triangle, with 42% being under 24 years old and an average age of 27.7 years (Canales et al., 2016). The importance of external migration, particularly rural migration, is evident in the remittance figures, as this population benefits from 51.3% of remittances in regions such as San Marcos, Huehuetenango, Quetzaltenango, Quiché, and Escuintla (International Organization for Migration, 2022b).

In this regard, the presented elements reveal that youth migration is not driven by a single factor but is rather complex and multi-causal in both countries. Additionally, shortcomings of both public and private actors responsible for creating optimal conditions for study, work, and safety at the local, regional, and national levels are evident (as explored in detail in the third section). It is noteworthy that the addressed scenarios do not differ much from other Latin American contexts, as a study conducted in Ecuador, Mexico, and Peru found that the poles of expulsion are concentrated in rural areas, particularly those with high levels of poverty and dependence on the agricultural sector (Cazzuffi and Fernández, 2018). A similar situation exists in Spain, where rural depopulation occurred in past decades, and currently, there is a phenomenon of youth migration from small cities to capitals like Madrid and Barcelona (González et al., 2022). Migration is not only a problem but also has effects on economic sectors, necessitating the implementation of public policies for mitigation. This will be further addressed in the following sections.

Similarly, it is worth noting that, as proposed by the *new economics of labor migration* (Gheasi and Nijkamp, 2017), youth migration is not solely an individual decision but part of a collective decision by families seeking better opportunities, or they are those who play the role of "chosen ones" to migrate on behalf of the household and overcome market deficiencies such as lack of employment. It is also evident that, according to world-system theory (Morawska, 2021), Guatemala and Colombia fail to meet the economic and social needs of their citizens, leading to a dependency on labor and educational opportunities from developed countries like the United States, Spain, and even Mexico. As a positive factor, the inflow of remittances into disadvantaged rural areas allows families to improve their quality of

TABLE 1 (Continued)



life, an aspect that aligns with the concepts of migration systems theory by linking migration to the transformation of social and economic conditions (macro factors) in the place of origin, thereby fostering development (Gheasi and Nijkamp, 2017). In this regard, the opposition between theories is noteworthy, as such development entails a level of dependency on wealthy countries, as expressed earlier.

4.2 Effects of migration on the agricultural sector

As described in the previous section, certain factors drive young people to leave rural areas and migrate to urban areas or abroad. This phenomenon impacts various economic sectors, including agriculture, and particularly cattle farming in the targeted countries. The following discusses these impacts, along with some characteristics of the sector.

In Colombia, the agricultural sector employs 15.2% of young people aged between 15 and 28 years (DANE, 2023a). The average salary is COP766957 per month (~US\$196) (DANE, 2022c), and informality reaches 85.4% (DANE, 2022d). It is noteworthy that departments with the highest cattle inventory, such as Antioquia (10.8%), Córdoba (8.3%), Casanare (8.2%), Meta (7.9%), Caquetá (6%), and Magdalena (5.7%) (DANE, 2022b), also present a high

percentage of youth migration. Antioquia is one of the regions with the highest youth migration (7.3%), followed by Meta (3.1%), Córdoba (2.7%), Magdalena (2.4%), Caquetá (1.6%), and Casanare (1.3%) (International Organization for Migration, 2021).

In these departments, and others with a focus on cattle farming, there is a loss of young labor, as seen in milk production systems in the Cañón de Anaime (Martínez et al., 2022). For producers, especially in medium-scale farms, retaining personnel for activities like milking is challenging (Ministerio de Agricultura y Desarrollo Rural, 2022a,b,c). Additionally, the sector demonstrates a low generational transfer, with an aging workforce and an inverted population pyramid (Ministerio de Agricultura y Desarrollo Rural, 2022a,b,c). According to the Third National Agricultural Census, the most recent and comprehensive statistic information at the national level, out of 725,225 cattle farmers living in rural areas (63.6% men and 36.4% women), the age group with the highest participation is individuals between 40 and 54 years old, constituting 32.7% (21.0% men and 11.7% women) (DANE, 2014). This situation negatively impacts the modernization of the sector, as older individuals lack sufficient knowledge of handling new technologies (Aguilar and Serrano, 2015) and often have a limited time horizon for long-term investments, i.e., in climate change adaptation and mitigation measures (Triana and Burkart, 2023; Figure 3).



In Guatemala, although the agricultural sector is the main source of employment, employing 35% of young people aged 15-29, it does not offer enough quality and well-paid jobs. The average salary is 1,155 quetzales per month (~US\$150), and 90% of the employed population is informal (FAO, 2020). Petén, with approximately 54% of the cattle inventory, is the department with the highest cattle population (Ministerio de Agricultura, Pesca y Alimentación, 2023). In order of importance, it is followed by Izabal (also in the north of the country), Jalapa, Jutiapa, Santa Rosa (southeast), Retalhuleu, and San Marcos (southwest) (Gobierno de la República de Guatemala, 2018). It is worth noting that Petén has a low participation rate in youth migration at the national level (3.71%), but San Marcos is one of the departments with the highest rates (22.52%). Other agricultural regions with youth migration include Quiché (12.50%) and Quetzaltenango (6.29%) (Villeda, 2020), although these areas do not have a strong focus on cattle farming.

In a similar fashion to Colombia, youth migration in Guatemala, predominantly by men, leads to a reduced availability of rural labor (Consejo Agropecuario Centroamericano, 2022). This is compounded by an increasing need for labor in the construction and mining sectors, which offer more attractive salaries (Weller, 2016). However, 50% of young people aged 15–24 live in rural areas, indicating that generational turnover is not yet a problem (Comisión Económica para América Latina y el Caribe, 2019). On the contrary, cattle-focused departments such as Petén, Izabal, and San Marcos have average ages of 24.36, 26.05, and 25.54 years, respectively (Instituto Nacional de Estadística (INE), 2018). This circumstance is crucial for the modernization of the sector, especially considering cultural barriers to digital literacy in the adult population. For

example, adults face challenges in understanding and using new technologies, and there is resistance to being instructed by children and young people on their use (Comisión Económica para América Latina y el Caribe, 2021).

In conclusion, youth migration emerges as an international issue, but the precarious labor conditions in local agricultural and livestock sectors in Colombia and Guatemala complicate the impacts, such as the availability of labor. In this regard, contrasting with the migration systems theory (Gheasi and Nijkamp, 2017), migration is not portrayed in positive terms linked to development, but rather as a phenomenon that threatens the livestock sector. Colombia's response to this issue through Venezuelan migration is significant, considering the country's hosting of 2,894,593 Venezuelan nationals. While the majority settle in urban areas (Grupo Interagencial sobre Flujos Migratorios Mixtos, 2023a), 345,739 are found in Antioquia alone (Grupo Interagencial sobre Flujos Migratorios Mixtos, 2023b), exemplifying the impact in cattle-focused regions and contributing to the labor force in this sector. However, the situation is different for Guatemala, as it is not a recipient country for migrants. Moreover, employing foreign labor does not provide a fundamental solution because it does not address the root causes of migration for the national population. Additionally, it is crucial to highlight that new technologies are an integral part of the cattle sector, including the use of robotics in animal husbandry or electronic data processing (Groher et al., 2020). Therefore, without a generational turnover capable of adapting to technological changes, these sectors in both countries will lag their international competitors. The above will affect job creation and the economy in general, perpetuating economic and social dependence on richer countries, as proposed by the world-system theory (Morawska, 2021).

4.3 Public policies to counteract youth migration

In the previous sections, the *push-pull* factors of youth migration were presented, along with their impacts on the agricultural, and particularly the cattle sector. Now, it is relevant to inquire about how governments have addressed this issue through the implementation of public policies and whether they have achieved the proposed results. To do this, it is pertinent to refer to the stages outlined by the KMPC.

In Colombia, the agenda setting for public policies has determined topics of interest such as job creation, study opportunities, and security for rural youth. This has led to a *design phase* addressing these issues (Figure 4; Table 2). During the period under consideration, one noteworthy policy is the document CONPES 173 of 2014, which considers the principles of Law 1,622 of 2013, known as the Youth Citizenship Statute (Consejo Nacional de Política Económica y Social, 2014). Among its objectives, the document aims to strengthen access to productive entrepreneurship programs and provide opportunities for human development that facilitate generational turnover in rural areas (Consejo Nacional de Política Económica y Social, 2014). The Law 1876 of 2017 emphasizes the need to involve young people and women in agricultural production, enabling collective and efficient management of inputs, food, raw materials, and other resources (Congreso de la República de Colombia, 2017). It is also noteworthy to mention the Productive Alliances for Life, which aim to connect small producers with large markets, giving preference in its calls for proposals to projects composed of a minimum of 51% women and/or men aged between 18 and 28 years (Ministerio de Agricultura y Desarrollo Rural, 2022a,b,c). Projects like the Comprehensive Agriculture and Rural Development Projects (PIDAR) follow similar selection criteria, seeking to encourage income generation for rural residents and enhance the country's competitiveness through the co-financing of projects (Agencia de Desarrollo Rural, 2022).

Regarding the lack of study opportunities, the Rural Youth Access to Higher Education program was strengthened since the National

Development Plan 2014–2018. This initiative provided credits and subsidies for academic program payments (Consejo Nacional de Política Económica y Social, 2021). The Generation E strategy also stands out, ensuring free enrollment for 200,000 young people in technical, technological, and university programs during its first 2 years of implementation (Ministerio de Educación Nacional, 2021). Among these, 63,000 students came from 719 municipalities considered rural or sparsely populated rural areas, and 54% of the beneficiaries were women (Consejo Nacional de Política Económica y Social, 2021). The promotion of agricultural and cattle technical education and double certification (academic-technical) in partnership with the National Learning Service (*Servicio Nacional de Aprendizaje*, SENA) has also been strengthened (FAO and SENA, 2021).

Similarly, the *agenda setting* and *design* phases of public policies in Guatemala reveal a comprehensive set of training policies aimed at mitigating the unemployment of rural youth (Figure 5; Table 3). One of the most prominent is the National Youth Policy 2012–2020, which promotes training and job placement processes through a generational, gender, and regional approach (Presidencia de la República de Guatemala, 2012). This initiative is aligned with the Agricultural Policy 2016–2020, which recognizes the issue of youth migration while proposing a boost to new information technologies and income diversification to generate dignified employment for young people, attracting them to agricultural activities (Ministerio de Agricultura, Ganadería y Alimentación, 2016).

With a greater emphasis on this population, the Rural Youth Strategy was formulated in 2017, among whose specific objectives is an offer of extension services that allow for the development of productive ventures (Ministerio de Agricultura, Ganadería y Alimentación, 2017). In the same year, the National Policy for Decent Employment 2017–2032 was presented, proposing the National Migration Program for Development as a priority action, fostering employment through the utilization of the "known how" of Guatemalan migrants with extended periods abroad (Ministerio de Trabajo y Previsión Social, 2017). Among other policies, it is essential



TABLE 2 Public policies to counteract youth migration in Colombia.

	Public policy	Promoting entity	Contribution of public policy
2013	Law 1,622. Youth Citizenship Statute (<i>Estatuto de Ciudadanía Juvenil</i>)	Congress of the Republic (<i>Congreso de la República</i>)	To ensure young people the full exercise of citizenship in civil, social, and public spheres
2014	CONPES 173. Guidelines for generating opportunities for youth (<i>Lineamientos para la generación de oportunidades</i> <i>para los jóvenes</i>)	National Council of Economic and Social Policy (<i>Consejo Nacional de Política Económica</i> <i>y Social</i>)	Develop strategies to ensure the transition of young people to working life in conditions of quality and stability
2014	Rural Youth Access to Higher Education (Jóvenes Rurales Acceso a la Educación Superior)	Ministry of Agriculture and Rural Development (Ministerio de Agricultura y Desarrollo Rural)	Improve access to rural education through credits and subsidies
2016	Final Agreement for the Termination of the Conflict (Acuerdo Final para la Terminación del Conflicto)	National Government; Revolutionary Armed Forces of Colombia (FARC-EP)[Gobierno Nacional; Fuerzas Armadas Revolucionarias de Colombia – Ejército del Pueblo (FARC-EP)]	End the internal armed conflict
2017	Law 1876. Creates the National Agri-food Innovation System (Crea el Sistema Nacional de Innovación Agropecuaria)	Congress of the Republic (<i>Congreso de la República</i>)	Ensure the participation of rural women and youth in processes offered by the National Agricultural Innovation System
2019	Generation E (Generación E)	National Government (Gobierno Nacional)	Increase access to quality higher education
2020	Productive Alliances for Life (Alianzas Productivas para la Vida)	Ministry of Agriculture and Rural Development (Ministerio de Agricultura y Desarrollo Rural)	Mitigate youth migration from rural areas to cities and strengthen income generation for young people and women
2021	Law 2,136. Comprehensive Migration Policy (<i>Política</i> Integral Migratoria)	Congress of the Republic (Congreso de la República)	Promote safe and orderly migration
2022	Comprehensive Agricultural and Rural Development Projects [Proyectos Integrales de Desarrollo Agropecuario y Rural (PIDAR)]	Rural Development Agency (Agencia de Desarrollo Rural)	Improve living conditions for rural residents, particularly women, youth, and conflict victime

Source: own elaboration.



to mention the Rural Learning Centers (CADER), driven by MAGA to transmit knowledge from a horizontal and practical approach in the quest for improving peasant economies (Asociación de Investigación y Estudios Sociales, 2022).

It is noteworthy that although violence is not one of the main push factors for migration, the Colombia's Peace Agreement between

FARC-EP and the government promotes the involvement of academic institutions in rural development. It also emphasizes promoting the productive permanence of young people in the field, beyond ending the armed confrontation between the State and the FARC-EP (Gobierno Nacional de Colombia y Fuerzas Armadas Revolucionarias de Colombia, 2016). In Guatemala, the National Security Policy TABLE 3 Public policies to counteract youth migration in Guatemala.

Public	policy	Promoting entity	Contribution of public policy
2012	National Youth Policy 2012–2020 (Política Nacional de Juventud 2012–2020)	Ministry of Social Development; National Youth Council; Planning and Programming Secretariat of the Presidency (<i>Ministerio de Desarrollo Social; Consejo Nacional de la Juventud; Secretaría de Planificación y</i> <i>Programación de la Presidencia</i>)	Improving the conditions and quality of life for young people by encouraging their integral development and citizenship exercise
2015	Training of Young Farmers (Formación de jóvenes Agricultores)	Helvetas (Swiss NGO); Ministry of Agriculture, Livestock, and Nutrition [Ministerio de Agricultura, Ganadería y Alimentación (MAGA)]	Training young people as entrepreneurs in the agricultural sector
2016	Decree 44. Migration Code (Código de Migración)	Congress of the Republic (Congreso de la República)	Recognizing the right of every person to migrate or emigrate
	Agricultural Policy 2016–2020 (Política Agropecuaria 2016–2020)	Ministry of Agriculture, Livestock, and Nutrition [Ministerio de Agricultura, Ganadería y Alimentación (MAGA)]	Strengthening the productivity of family farmers with an emphasis on young people and women
	Youth Agricultural Development Centers (CADER Juveniles)	Ministry of Agriculture, Livestock, and Nutrition [Ministerio de Agricultura, Ganadería y Alimentación (MAGA)]	Increasing the productivity of rural families
2017	National Security Policy (Política Nacional de Seguridad)	National Security Council (Consejo Nacional de Seguridad)	Proposing comprehensive actions for national security
	National Policy for Decent Employment 2017–2032 (Política Nacional de Empleo Digno 2017–2032)	Ministry of Labor and Social Welfare; International Labor Organization for Central America (Ministerio de Trabajo y Previsión Social; Organización Internacional del Trabajo para América Central)	Reducing poverty and inequality among young people and women
	Strategy for Rural Youth (Estrategia para la Juventud Rural)	Ministry of Agriculture, Livestock, and Nutrition [Ministerio de Agricultura, Ganadería y Alimentación (MAGA)]	Promoting equity and social inclusion for rural youth
2018	National Strategy for Sustainable Low Emission Cattle in Guatemala (Estrategia Nacional de Ganadería Bovina Sostenible con Bajas Emisiones de Guatemala)	Public sector; private sector; academia; NGOs; cattle sector organizations	Greater inclusion of young people and women in cattle organizations

Source: own elaboration.

proposes a local governance program where security is addressed jointly with other social needs, such as job creation (Consejo Nacional de Seguridad, 2017).

Other policies addressing various migration causes adhere to these policies. In the Decree 44 of 2016, Guatemala recognizes that its social and economic conditions generate a high number of migrants, necessitating the regulation of the phenomenon. Specifically for young people, it establishes the need for authorizations from parents for their departure, as well as compliance with the requirements of the receiving country (Congreso de la República de Guatemala, 2016). In Colombia, the Law 2,136 of 2021 acknowledges a diaspora of citizens abroad due to a multifactorial reality, aiming to promote orderly migration (Congreso de la República de Colombia, 2021). It establishes requirements such as the need for minors to have an exit permit from their father or mother, as stipulated by Article 110 of Law 1,098 of 2006 (Congreso de la República de Colombia, 2021).

Regarding policies promoted by the cattle sector, Guatemala's Sustainable Cattle National Strategy proposes strengthening the inclusion of young people and women in productive and industrial organizations related to dairy and beef production. It also aims to increase the training of young technicians (Gobierno de la República de Guatemala, 2018). In contrast, Colombia's Sustainable Cattle Policy, while referring to job creation, does not explicitly address this population (Ministerio de Agricultura y Desarrollo Rural, 2022a,b,c). In both scenarios, parallel to public policies, initiatives have been developed by the private sector. In Guatemala, it is crucial to refer to the Youth Farmers Agro-entrepreneurs Training Program [Formación de Jóvenes Agricultores Agro-empresarios (FORJA)], which prioritizes young people excluded from formal education systems (Ministerio de Agricultura, Ganadería y Alimentación, 2017). The program has been implemented in the departments of San Marcos, Huehuetenango, Alta Verapaz, and Quetzaltenango, training 985 young people, with 36% being women (Ministerio de Agricultura, Ganadería y Alimentación, 2017). In Colombia, the Heirs of Tradition program (Herederos de Tradición), driven by Alquería and SENA, aimed to engage rural youth in technological programs for cattle production. The program trained around 189 young people (60 women and 129 men) in five promotions, particularly in the departments of Meta, Cundinamarca,

Nariño, Santander, and Cesar (Triana et al., 2020; Triana and Burkart, 2023).

It is emphasized that the *adoption* of policies in both Colombia and Guatemala does not reveal opposition or veto players; on the contrary, there is a joint effort between the legislative and executive branches, involving multiple supportive actors, such as National Governments and Ministries, among others. However, the *implementation* of these policies shows an exposition of objectives without clear long-term budgets to ensure sustainability. Furthermore, after a decade of implementation, the *evaluation* is not entirely favorable (Figure 6).

In Guatemala, while urban youth aged 19–24 have an average of 9.10 years of education, rural youth in the same range only have 6.64 years (Consejo Nacional de la Juventud y Fondo de Población de las Naciones Unidas, 2020). Regarding gender, there are slight differences for the same age group. From the total population (48.3% men and 51.7% women), basic education was completed by 14.8% of men and 12.1% of women (Consejo Nacional de la Juventud y Fondo de Población de las Naciones Unidas, 2020). In Colombia, a stark contrast exists between urban and rural areas. While the average duration of study in urban areas is 9.2 years, in rural areas, it is only 5.5. Additionally, only two out of 10 rural high school graduates manage to enter higher education (Ministerio de Agricultura, Ganadería y Alimentación, 2017). However, unlike Guatemala, the percentage of rural youth (20–24 years) completing secondary

education is higher in women (44.63%) than in men (38.64%) (Comisión Económica para América Latina y el Caribe, 2019).

Similarly, employment policies in Colombia reveal negative results, as the national youth unemployment rate (15-28 years) is 19% (DANE, 2023b). Unemployment rates are also unfavorable in cattle departments such as Córdoba (12.5%), Meta (11.2%), Caquetá (11.2%), and Magdalena (9.8%). These rates do not differ significantly from departments receiving migrants, and the latter even have higher figures, i.e., Antioquia (10.1%), Cundinamarca (12.7%), and Valle del Cauca (13.2%) (DANE, 2023c). In contrast, Guatemala has an unemployment rate of only 2.5%, but underemployment affects 9.4% of youth (15-24 years). Furthermore, 68% of the unemployed are under 30 years old, with those between 20 and 24 years being the most affected at 33.1% (Instituto de Investigaciones Económicas y Sociales, 2023). It is also noteworthy that the department of Guatemala has one of the highest percentages of wage employment at 60%, while agricultural and/or rural areas like Huehuetenango and Jalapa only have 34.9 and 35.1%, respectively (Programa de las Naciones Unidas para el Desarrollo, 2022).

These scenarios align with the situation in Mexico, where despite the objectives outlined in policies, they present partial results with moderate impact and no structural solutions (García et al., 2019). It is also worth noting that, while migration statutes contribute by regulating the phenomenon, in neither of the two countries of analysis do they aim to counteract its causes. Moreover, the Sustainable Cattle



Policy in Colombia does not address rural youth employment, in contrast to the one in Guatemala. Similarly, while strategies initiated by the private sector are important, it should be the State leading employment and training programs in rural areas and not relinquishing this function. The pursuit of gender parity is a constant in the policies discussed, but results in this regard are not fully satisfactory, with men being the main participants and beneficiaries. This situation is not exclusive to Colombia and Guatemala but aligns with a Latin American context where legislative amendments and positive discrimination policies have not led to coherence between objectives and results, persisting gaps in access to credit, training, and technical assistance, among others (Valenciano et al., 2016).

Regarding the theories of migration addressed, several aspects stand out. Firstly, the failure of public policies and their inability to resolve the push factors of migration in Colombia and Guatemala have not allowed overcoming the structural dependence mentioned by the *world-system theory*. Instead, wealthy countries like the United States and Spain absorb their young workforce. Similarly, it is revealed that better social conditions and quality of life have not been generated to encourage the return of migrants, which is a possibility from the perspective of the *new economics of labor migration*. In contrast, migration continues to be a risk management strategy for young people and their families. Finally, besides the negative effects highlighted for the cattle sector, both countries lack mechanisms to maximize the positive impacts of migration mentioned by the *migration systems theory*, such as providing financial education to families to make effective use of remittances.

5 Conclusion and recommendations

Considering the research question, an evident conclusion unveils that unemployment, lack of study opportunities, and insecurity are the main drivers of rural youth migration in Colombia and Guatemala. This phenomenon has impacted the cattle sector by creating a shortage of labor and hindering generational transfer, thereby causing difficulties for the sector's modernization, as well as implementing climate change adaptation and mitigation measures. Although multiple policies have been implemented to counteract these drivers, the results are negative, and the issue persists. In this regard, if the national governments in both countries do not adopt policies that go beyond partial responses, the cattle sector will lag behind its regional and international competitors, generating negative impacts on the achievement of the Sustainable Development Goals, i.e., SDG-1 (no poverty), SDG-2 (zero hunger), SDG-6 (gender equality), SDG-8 (decent work and economic growth), SDG-10 (reduced inequalities), SDG-13 (climate action), among others. These policies must continue while preserving their positive aspects, such as gender and youth perspectives, but with greater investments to reach a broader beneficiary population. Although this is a common consideration in public policy analysis, it is necessary to reiterate it because budgets are a crucial condition for achieving the stated objectives.

On the other hand, it is relevant to discuss the relevance of the theoretical models used and their potential contributions. The *push-pull theory of migration* is considered broad and flexible enough, allowing the exploration of a variety of variables (employment, education, security, etc.) and, thus, a comprehensive understanding of the phenomenon. It facilitates addressing migration in both its individual causes (unemployed youth) and structural causes (underfunded public policies). A theoretical contribution is seen in the proposed link

between migration and public policies, as well as the impacts of the issue on a specific economic sector, such as cattle farming. Also highlighted is the contrast between the exposed negative effects of migration and the *migration systems theory*, suggesting a need to revise the positive conception of the phenomenon proposed by the theory. Regarding the *Kaleidoscope Model of Policy Change*, its capacity to understand public policies as a process that goes beyond setting objectives or executing solutions, requiring continuous evaluation, is reaffirmed. As a primary contribution to this theoretical model, its application in Latin American contexts stands out, as its antecedents are primarily in African countries. Likewise, it stands out that the lack of a specific theory to address the migration-cattle relationship was overcome by articulating various theoretical postulates, thus constructing an analytical framework that can be used in future research.

Among the methodological recommendations, there is a proposal to expand local-level studies addressing the links between youth migration and generational transfer, lack of labor, and modernization of the cattle sector, as research on this aspect is considered insufficient in the contexts addressed. Additionally, it is recommended to conduct comparative studies between developing countries (such as Colombia and Guatemala) and developed countries (such as the United States and European nations) with the aim of identifying successes, failures, and commonalities among implemented policies. Another possibility is the development of studies that contrast countries with and without internal armed conflict, which would allow identifying differences between migration due to violence and migration due to social or economic reasons.

In terms of practical recommendations, there is an emphasis on the need for Colombia's Sustainable Cattle Policy to focus on social components for generating employment for youth and women. For both countries, the design of a policy that aims to counteract the issue rather than merely regulating migration is suggested. Finally, it is proposed to establish financial education programs for families receiving remittances so that they can develop microenterprises that promote employment and prevent future migration.

Data availability statement

Data will be made available upon reasonable request.

Author contributions

MD: Conceptualization, Formal analysis, Investigation, Methodology, Resources, Writing – original draft, Writing – review & editing. LM: Formal analysis, Resources, Writing – original draft, Writing – review & editing. SB: Conceptualization, Funding acquisition, Project administration, Resources, Supervision, Writing – original draft, Writing – review & editing. NT: Funding acquisition, Methodology, Project administration, Supervision, Writing – original draft, Writing – review & editing.

Funding

The author(s) declare that financial support was received for the research, authorship, and/or publication of this article. This work was

funded by the OneCGIAR Initiative Livestock & Climate (L&C). The funders had no role in the design of the study in the collection, analyses, or interpretation of data; in the writing of the manuscript, or in the decision to publish the results.

Acknowledgments

We thank all donors who globally support our work through their contributions to the CGIAR System.

Conflict of interest

The authors declare that the research was conducted in the absence of any commercial or financial relationships that could be construed as a potential conflict of interest.

References

Agencia de Desarrollo Rural (2022). Acuerdo 005. Por el cual se definen los criterios de calificación de los Proyectos Integrales de Desarrollo Agropecuario y Rural con Enfoque Territorial (PIDAR) para la vigencia 2022. Bogotá. Available at: https://www. adr.gov.co/wp-content/uploads/2022/09/Acuerdo-No.-005-de-2020.-Criterios-de-calificacio%E2%95%A0un-de-PIDAR-Vigencia-2022.pdf

Aguilar, T., and Serrano, D. (2015). Analysis of the social component of livestock in a northern region of Colombia (municipality of Montería). *Perspect. Geogr.* 20, 133–154. doi: 10.19053/01233769.4502

Aguirre, K. (2014). Analyzing violence after the conflict: Guatemala as a case of a subnational study. Rev. Mex. Cienc. Políti. Soc. 59, 191–234. doi: 10.1016/S0185-1918(14)70805-0

Asociación de Investigación y Estudios Sociales (2022). El aporte del SNER y de los CADER al desarrollo rural en Guatemala. Guatemala Available at: http://www.asies.org. gt/download.php?get=el_aporte_del_sner_y_de_los_cader_al_desarrollo_rural_en_guatemala.pdf.

Barrios, M., Cabello, J., Calva, L., Carrión, V., Cruz, R., Cueva, T., et al. (2022). Migración centroamericana: factores de expulsión y factores de atracción de la población guatemalteca, salvadoreña y hondureña desde la perspectiva de género. Ciudad de México: Centro Global de Excelencia en Estadísticas de Género y Colegio de la Frontera Norte Available at: https://mexico.un.org/sites/default/files/2023-02/ONUMX_ Migracion_Completo_Enero_digital-V3.pdf.

Canales, A., Fuentes, J., and León, C. (2016). Desarrollo y migración. Desafíos y oportunidades en los países del norte de Centroamérica. Santiago de Chile, Chile: Comisión Económica para América Latina y el Caribe (CEPAL), Organización de las Naciones Unidas para la Alimentación y la Agricultura (FAO).

Castelli, F. (2018). Drivers of migration: why do people move? J. Travel Med. 25, 1–7. doi: 10.1093/jtm/tay040

Cazzuffi, C., and Fernández, J. (2018). Rural youth and migration in Ecuador, Mexico and Peru. Santiago de Chile: Latin American Center for Rural Development Available at: https://www.rimisp.org/wp-content/files_mf/1539440301DocumentoTrabajoenproc eso_RuralYouthMigration_Cazzuffi_Fernandez_2018.pdf.

Chacón, E. (2021). Proposal to measure and categorize rurality in Costa Rica. Perspect. Rural. Nueva Época 18, 1–21. doi: 10.15359/prne.18-36.1

Comisión Económica para América Latina y el Caribe (2019). Situación de las juventudes rurales en América Latina y el Caribe. Ciudad de México. Available at: https://repositorio.cepal.org/server/api/core/bitstreams/f94f1075-5e5a-4aa1-aa1d-c203d026afce/content

Comisión Económica para América Latina y el Caribe (2021). Análisis de las políticas públicas e iniciativas privadas que apoyan el uso de las tecnologías digitales en las mipymes agrícolas y agroindustriales en Guatemala. Santiago de Chile Available at: https://repositorio.cepal.org/server/api/core/bitstreams/0d550ea3C25e7-49bd-bfeb-13da9e7d6f5a/content.

Congreso de la República de Colombia (2017). Ley 1876. Por medio de la cual se crea el Sistema Nacional de Innovación Agropecuaria y se dictan otras disposiciones. Bogotá Available at: http://www.secretariasenado.gov.co/senado/basedoc/ ley 1876 2017.html.

Congreso de la República de Colombia (2018). Por Ley Estatutaria 1885. la cual se modifica la Ley Estatutaria 1622 de 2013 y se dictan otras disposiciones. Bogotá. Available at: https://www.suin-juriscol.gov.co/viewDocument.asp?id=30034547

Congreso de la República de Colombia (2021). Ley 2136. Por medio de la cual se establecen las definiciones, principios y lineamientos para la reglamentación y

Publisher's note

All claims expressed in this article are solely those of the authors and do not necessarily represent those of their affiliated organizations, or those of the publisher, the editors and the reviewers. Any product that may be evaluated in this article, or claim that may be made by its manufacturer, is not guaranteed or endorsed by the publisher.

Author disclaimer

The views expressed in this document may not be taken as the official views of these organizations. CGIAR is a global research partnership for a food-secure future. Its science is carried out by 15 Research Centres in close collaboration with hundreds of partners across the globe.

orientación de la política integral migratoria del Estado colombiano (PIM) y se dictan otras disposiciones. Bogotá Available at: https://www.alcaldiabogota.gov.co/sisjur/normas/Norma1.jsp?i=115641

Congreso de la República de Guatemala (2016). "Código de Migración" in Decreto 44–2016 del Congreso de la República (Ciudad de: Guatemala). Available at: https://igm.gob.gt/wp-content/uploads/2021/11/1_Codigo-de-Migracion-Decreto-44-2016-del-Congreso-de-la-Republica.pdf

Consejo Agropecuario Centroamericano (2022). Juventudes Rurales de la Región SICA. Available at: https://www.cac.int/sites/default/files/Est%20Juventudes%20 Rurales%20de%20la%20Regio%CC%81n%20SICA%202022-2030_Final.pdf

Consejo Nacional de la Juventud y Fondo de Población de las Naciones Unidas (2020). Juventudes en Guatemala. Guatemala: Documento analítico de juventudes en Guatemala Available at: https://guatemala.unfpa.org/sites/default/files/pub-pdf/documento_ analitico_final.pdf.

Consejo Nacional de Política Económica y Social (2014). Documento CONPES 173. Lineamientos para la generación de oportunidades para los jóvenes. Bogotá Available at: https://colaboracion.dnp.gov.co/CDT/Conpes/Social/173.pdf.

Consejo Nacional de Política Económica y Social (2021). Documento CONPES 4040. Pacto Colombia con las juventudes: estrategia para fortalecer el desarrollo integral de la juventud. Bogotá. Available at: https://colaboracion.dnp.gov.co/CDT/Conpes/ Econ%C3%B3micos/4040.pdf

Consejo Nacional de Seguridad (2017). Política Nacional de Seguridad. Ciudad de Guatemala. Available at: https://stcns.gob.gt/wp-content/uploads/2022/08/Politica-Nacional-deSeguridad-2017-modif-07may2021_compressed.pdf

DANE (2014). Censo Nacional Agropecuario. Bogotá: Caracterización de los productores residentes en el área rural dispersa censada Available at: https://www.dane.gov.co/files/CensoAgropecuario/entrega-definitiva/Boletin-2-Productores-residentes/2-Boletin.pdf.

DANE (2022a). Reporte Estadístico de Migración. Bogotá. Available at: https:// www.dane.gov.co/files/investigaciones/poblacion/migracion/2doreporte-migracion. pdf

DANE (2022b). Caracterización de la actividad ganadera a partir del aprovechamiento de registros administrativos. Bogotá. Available at: https://n9.cl/2fem3

DANE (2022c). Perspectivas del sector agropecuario en Colombia. Bogotá. Available at: https://www.dane.gov.co/files/investigaciones/planes-departamentosciudades/220331-Presentacion-general-Asoleche-fin.pdf

DANE (2022d). Mercado laboral. Bogotá: Nueva medición de informalidad laboral Available at: https://www.dane.gov.co/files/investigaciones/boletines/ech/ech/Nueva_ medicion_informalidad.pdf.

DANE (2023a). Mercado laboral de la juventud. Bogotá. Available at: https://www. dane.gov.co/files/investigaciones/boletines/ech/juventud/boletin_GEIH_juventud_ sep22_nov22.pdf

DANE (2023b). Mercado laboral de la juventud. Bogotá. Available at: https://www. dane.gov.co/files/investigaciones/boletines/ech/juventud/boletin_GEIH_juventud_ ene23_mar23.pdf

DANE (2023c). Mercado laboral por departamentos. Bogotá. Available at: https:// www.dane.gov.co/files/investigaciones/boletines/ech/ml_depto/Boletin_dep_22.pdf

Datos Macro (2020). Guatemala. Emigrantes totales. Available at: https://datosmacro.expansion.com/demografia/migracion/emigracion/guatemala

Díaz, M., Moreno, L., and Burkart, S. (2024). How do sustainability policies emerge in the Colombian political system? A kaleidoscope model analysis of the policy for sustainable cattle 2022–2050. *Clean Circ. Bioecon.* 7:100075. doi: 10.1016/j. clcb.2024.100075

Everaert, I. (2021). Los procesos migratorios en la incidencia de la violencia contra las mujeres y niñas: Programa de las Naciones Unidas para el Desarrollo (PNUD) Available at: https://oig.cepal.org/sites/default/files/e1_informe_final_ccrev_07082022.pdf.

Fangqu, N. (2022). A push-pull model for inter-city migration simulation. *Cities* 131:104005. doi: 10.1016/j.cities.2022.104005

FAO (2018). World livestock: transforming the livestock sector through the sustainable development goals. Rome.

FAO (2020). Empleo rural juvenil y sistemas agroalimentarios en Guatemala. Available at: https://www.fao.org/3/ca8062es/CA8062ES.pdf

FAO and SENA (2021). Escuela Nacional de Extensión Rural: Proyecto Educativo Available at: https://www.sena.edu.co/es-co/Noticias/Documents/Cartilla_ ENER_14052021.pdf.

García, A. (2018). Revisión crítica de las principales teorías que tratan de explicar la migración. *España* 7, 198–228. doi: 10.25115/riem.v7i4.1963

García, V., Masferrer, C., and Giorguli, S. (2019). Cambios futuros en la estructura etaria y diferentes escenarios migratorios. El caso de América del Norte y Central. *Rev. Latinoam. Pobl.* 13, 36–53. doi: 10.31406/relap2019.v13.i2.n25.2

Gaspar, S., and Chatti, J. (2022). Migración brasileña en Portugal y España: dinámicas y nuevos paradigmas. *Migraciones* 56, 1–19. doi: 10.14422/mig.2022.025

Gheasi, M., and Nijkamp, P. (2017). A brief overview of international migration motives and impacts, with specific reference to FDI. *Economies.* 5:31. doi: 10.3390/ economies5030031

Gobierno de la República de Guatemala (2018). Estrategia Nacional de Ganadería Bovina Sostenible con Bajas Emisiones. Guatemala Available at: https://www.maga.gob. gt/download/estrategiaganado.pdf.

Gobierno Nacional de Colombia y Fuerzas Armadas Revolucionarias de Colombia (2016). Acuerdo Final para la Terminación del Conflicto y la Construcción de una Paz Estable y Duradera. La Habana. Available at: https://www.cancilleria.gov.co/sites/default/files/Fotos2016/12.11_1.2016nuevoacuerdofinal.pdf

González, M., López, A., and Esteve, A. (2022). Interregional migration of human capital in Spain. *Reg. Stud.* 9, 324–342. doi: 10.1080/21681376.2022.2060131

Groher, T., Heitkämper, K., and Umstätter, C. (2020). Digital technology adoption in livestock production with a special focus on ruminant farming. *Animal* 14, 2404–2413. doi: 10.1017/S1751731120001391

Grupo Interagencial sobre Flujos Migratorios Mixtos (2023a). Infografía de refugiados y migrantes venezolanos. Available at: https://www.r4v.info/sites/default/files/2023-03/ Infograf%C3%ADa_Refugiados_y_Migrantes_Costa_Caribe_27022023.pdf

Grupo Interagencial sobre Flujos Migratorios Mixtos (2023b). Afiliaciones a salud de personas refugiadas y migrantes venezolanas. Antioquia. Available at: https://reliefweb. int/report/colombia/afiliaciones-salud-de-personas-refugiadas-y-migrantes-venezolanas-gifmm-antioquia-septiembre-2023

Hass, H. (2021). A theory of migration: the aspirations-capabilities framework. Comp. Migr. Stud. 9:8. doi: 10.1186/s40878-020-00210-4

Hendriks, S., Babu, S., and Haggblade, S. (2017). What drives nutrition policy reform in Africa? Applying the kaleidoscope model of food security policy change. Feed the Future Innovation Lab for Food Security Policy Research Brief 30. East Lansing: Michigan State University. Available at: https://www.canr.msu.edu/resources/whatdrives-nutrition-policy-reform-in-africa

Howlett, M., and Cashore, B. (2014). "Conceptualizing Public Policy" in Comparative policy studies. eds. I. Engeli and C. Allison (London: Palgrave Macmillan), 17–33.

Instituto de Investigaciones Económicas y Sociales (2023). El mercado laboral en Guatemala a la luz de la ENEI 2021. *Guatemala* 235, 1–52.

Instituto Nacional de Estadística (INE) (2018). Resultados del Censo 2018. Ciudad de Guatemala. Available at: https://www.censopoblacion.gt/mapas

Inter-American Development Bank (2021). Migration flows in Latin America and the Caribbean: statistics on permits for migrants. Washington, D.C.: IDB, OECD.

International Organization for Migration (2020). Migration Trends in South America. Available at: https://robuenosaires.iom.int/sites/g/files/tmzbdl626/files/documents/ Migration_Trends_final.pdf

International Organization for Migration (2022a). Recent migration trends in the Americas. Available at: https://rosanjose.iom.int/sites/g/files/tmzbdl1446/files/ documents/recent-migration-trends-in-the-americas_en.pdf

International Organization for Migration (2022b). Encuesta sobre migración internacional de personas guatemaltecas y remesas 2022. Ciudad de Guatemala. Available at: https://infounitnca.iom.int/uploads/RemesasGT2022/Resumen_ EncuestaRemesas2022_gt.pdf

International Organization for Migration (2021). Perfil Migratorio de Colombia 2021. Colombia. Available at: https://publications.iom.int/system/files/pdf/MP-Colombia-2021-ES.pdf Jiménez, E. (2016). La violencia en el Triángulo Norte de Centroamérica: una realidad que genera desplazamiento. *Papel Polít.* 21, 167–196. doi: 10.11144/Javeriana.papo21-1. vtnc

López, E., Martínez, L., Martínez, C., and Prieto, A. (2017). Rural development and aging: case study in Chinavita, Boyacá, Colombia. *Rev. Invest. Desarr. Innova.* 8, 193–206. doi: 10.19053/20278306.v8.n2.2018.7959

Lozano, D., Chacón, A., Gutiérrez, I., and Robalino, J. (2015). Eventos climáticos extremos y migración interna en Guatemala, un análisis basado en percepciones de expertos. *Ciencia Ergo Sum.* 22, 35–44.

Lune, H., and Berg, B. (2017). Qualitative research methods for the social sciences. *9th* Edn: Pearson Available at: http://law.gtu.ge/wp-content/uploads/2017/02/Berg-B.-Lune-H.-2012.-Qualitative-Research-Methods-for-the-Social-Sciences.pdf.

Martínez, G., Menéndez, C., and Mora, J. (2022). Peasant families and milk production systems in the Anaime canyon (Colombia). *Perspect. Rural.* 21, 1–25. doi: 10.15359/ prne.21-41.5

Mather, D., and Ndyetabula, D. (2016). Assessing the drivers of Tanzania's fertilizer subsidy programs from 2003-2016: an application of the kaleidoscope model of policy change. Feed the Future, USAID: Michigan State University.

Mina, Y., and Téllez, G. (2022). Oportunidades laborales y la migración rural-urbana juvenil. Juventud rural araucana. *Estud. Polít.* 65, 89–118. doi: 10.17533/udea.espo. n65a03

Ministerio de Agricultura, Ganadería y Alimentación (2016). Política Agropecuaria 2016–2020. Ciudad de Guatemala. Available at: https://www.maga.gob.gt/download/politica-16-20.pdf

Ministerio de Agricultura, Ganadería y Alimentación (2017). Estrategia para la Juventud Rural. Ciudad de Guatemala. Available at: https://www.maga.gob.gt/download/estrategia-juventud.pdf

Ministerio de Agricultura, Pesca y Alimentación (2023). Sectores agroalimentario y pesquero. España. Available at: https://www.mapa.gob.es/es/ministerio/ministerio/america-central-caribe/fichasectores_gt-reduc_tcm30-578265.pdf

Ministerio de Agricultura y Desarrollo Rural (2022a). Plan de Ordenamiento Productivo. Bogotá: Cadena Láctea en Colombia Available at: https://www.andi.com.co/ Uploads/10_05_2022_POP_Cadena_La%CC%81ctea_Bovina.pdf.

Ministerio de Agricultura y Desarrollo Rural (2022b). Política de Ganadería Bovina Sostenible 2022–2050. Bogotá. Available at: https://www.minagricultura.gov.co/ Normatividad/Resoluciones/RESOLUCI%C3%93N%20NO.%20000126%20DE%20 2022.pdf

Ministerio de Agricultura y Desarrollo Rural (2022c). Términos de referencia para apertura del registro de perfiles de alianzas 2022. Bogotá. Available at: https://www. minagricultura.gov.co/tramites-servicios/desarrollo-rural/Documents/Terminos%20 de%20referencia%20TdR%20PAAP%202022%20Ene-24-2022%20.pdf

Ministerio de Educación Nacional (2021). A través del programa Generación E más de 200.000 jóvenes han accedido a Educación Superior: Bogotá Available at: https://www.mineducacion.gov.co/1780/articles-405367_txt_audio_1.pdf.

Ministerio de Trabajo y Previsión Social (2017). Política Nacional de Empleo Digno 2017–2032. Ciudad de Guatemala. Available at: https://www.mintrabajo.gob.gt/images/ejes/PNED_2017_2032.pdf

Mohamed, A., and Abdul, A. (2020). Push-pull factors influencing international return migration intentions: a systematic literature review. *J. Enterpr. Commu. People Places Glob. Econ.* 14, 231–246. doi: 10.1108/JEC-01-2020-0004

Morales, L., Santos, W., and Segura, E. (2015). The rural development: an approach from the new rurality. *Cienc. Agric.* 12, 15–25. doi: 10.19053/01228420.4110

Morawska, E. (2021). "Historical-structural models of international migration" in An introduction to international migration studies: European perspectives. ed. M. Martiniello (Amsterdam: Amsterdam University Press), 57–78.

Moreno, L., Díaz, M., and Burkart, S. (2022). Public policies for the development of a sustainable cattle sector in Colombia, Argentina, and Costa Rica: a comparative analysis (2010–2020). *Front. Sustain. Food Syst.* 6, 1–18. doi: 10.3389/fsufs.2022. 722522

OECD and FAO (2019). Perspectivas agrícolas OCDE-FAO 2019–2028. Enfoque Especial: América Latina Available at: https://n9.cl/m1jg9.

Palmatier, R., Houston, M., and Hulland, J. (2018). Review articles: purpose, process, and structure. J. Acad. Mark. Sci. 46, 1–5. doi: 10.1007/s11747-017-0563-4

Presidencia de la República de Guatemala (2012). Política Nacional de Juventud 2012–2020. Ciudad de Guatemala. Available at: https://conjuve.gob.gt/wp-content/uploads/2020/08/PNJ-2012-2020.pdf

Prieto, R., Quiñones, M., Lora, E., and O'Clery, N. (2022). Mobility between Colombian cities is predominantly repeat and return migration. *Comput. Environ. Urban. Syst.* 94, 101774–101713. doi: 10.1016/j.compenvurbsys.2022.101774

Programa de las Naciones Unidas para el Desarrollo (2022). Informe Nacional de Desarrollo Humano. Available at: https://indhguatemala.org/

Resnick, D., Haggblade, S., Babu, S., Hendriks, S., and Mather, D. (2018). The kaleidoscope model of policy change: applications to food security policy in Zambia. *World Dev.* 109, 101–120. doi: 10.1016/j.worlddev.2018.04.004

Rocha, J., and Gramajo, L. (2017). Migración reciente en el altiplano occidental guatemalteco: redes, reunificación familiar y efecto demostración. *Eutopía* 3, 3–42.

Rosendo, A., Herrera, F., Vizcarra, I., and Baca, N. (2019). Rural territorial development: agriculture and migration in the south of the state of Mexico. *Econ. Soc. Territ.* 19, 1243–1274. doi: 10.22136/est20191207

Sahu, P. (2013). "Research process" in Research methodology: a guide for researchers in agricultural science, social science and other related fields (New Delhi: Springer New Delhi), 15–20.

Sampieri, R., Fernández, C., and Baptista, M. (2014). Metodología de la investigación. México: McGraw Hill Available at: https://www.uca.ac.cr/wp-content/uploads/2017/10/ Investigacion.pdf.

Sandoval, D., Moctezuma, S., Herrera, F., and Espinoza, A. (2022). Rural youth: a perspective of agricultural labor from their protagonists. *Convergencia* 29, 1–23. doi: 10.29101/crcs.v29i0.16508

Santibáñez, J., Palma, S., Ramírez, H., Hidalgo, N., and Urban, A. (2017). Estudio cualitativo sobre mujeres jóvenes y la violencia en Centroamérica: efecto en condiciones de salida y retorno de menores migrantes. *Nota Técnica No IDB-TN-117*. doi: 10.18235/0000630

Schiavon, J. (2023). How should migration be studied in the meso-north American system? *Norteam. Rev. Acad. CISAN-UNAM* 17, 171–190. doi: 10.22201/ cisan.24487228e.2022.2.585

Sirbu, A., Andrienko, G., Andrienko, N., Boldrini, C., Giannotti, F., and Guidotti, R., et al. (2021). Human migration: the big data perspective. Int. J. Data Sci. Anal. 11, 341–360. doi: 10.1007/s41060-020-00213-5

Spanevello, R., Moreira, S., Matte, A., and Boscardin, M. (2022). La migración juvenil y las dificultades de reproducir la mano de obra en propiedades rurales familiares brasileñas. *Rev. Desarr. Econ. Territorial.* 22, 54–73. doi: 10.17141/eutopia.23.2022.5585

Spohn, M. (2017). Emigrants' stories of foreign aid and their reasons for emigration: Guatemalans on the move. Univ. Psicol. 16, 1–15. doi: 10.11144/Javeriana.upsy16-5.esfa

Statista (2023a). Número de emigrantes internacionales en países de América Latina en 2020. Available at: https://es.statista.com/estadisticas/1261872/numero-deemigrantes-internacionales-por-pais-de-latinoamerica/#:~:text=En%202020%20 M%C3%A9xico%20se%20posicion%C3%B3,cinco%20y%20tres%20millones%2C%20 respectivamente

Statista (2023b). Valor de remesas personales recibidas en países selectos de América Latina y el Caribe en, 2022 Available at: https://es.statista.com/estadisticas/1261886/ remesas-recibidas-por-pais-de-latinoamerica/#:~:text=M%C3%A9xico%20se%20 posicion%C3%B3%20como%20el,9.440%20millones%20de%20d%C3%B3lares%20 respectivamente. Suárez, M. (2023). Returned migrant children and youths in Mexico: findings, advances, and pending issues (2015-2022). *Norteamericana* 17, 241–265. doi: 10.22201/cisan.24487228e.2022.2.587

Toney, M., and Bailey, A. (2014). "Migración, una visión general" in Encyclopedia of quality of life and well-being research. ed. A. C. Michalos (Dordrecht: Springer Dordrecht), 4044–4050.

Torres, M. (2021). Migración en Colombia: causas, efectos y estrategias de retorno. *Arkitekturax Vis.* 3, 1–26. doi: 10.29097/26191709.303

Triana, N., Ariza, M., and Burkart, S. (2020). Juventud ganadera y el poder transformativo de la educación rural: el caso de Herederos de Tradición, 2012–2020. Agriculture and Food Security: CGIAR Research Program on Climate Change Available at: https://cgspace.cgiar.org/items/b9df23f9-173d-4353-adc8-042522a63c95.

Triana, N., and Burkart, S. (2023). Youth in livestock and the power of education: the case of heirs of tradition from Colombia, 2012–2020. *J. Rural. Stud.* 97, 405–415. doi: 10.1016/j.jrurstud.2022.12.032

Unidad de Víctimas (2017). Víctimas del conflicto armado. Available at: https://www.unidadvictimas.gov.co/es/registro-unico-de-victimas-ruv/

United Nations (2014). Juventud 2030. Trabajando con y para los jóvenes. Available at: https://www.un.org/youthenvoy/wp-content/uploads/2014/09/UN-Youth-Strategy-Spanish1.pdf

United Nations (2018). La Agenda 2030 y los Objetivos de Desarrollo Sostenible: una oportunidad para América Latina y el Caribe. Santiago Available at: https://repositorio. cepal.org/server/api/core/bitstreams/cb30a4de-7d87-4e79-8e7a-ad5279038718/content.

Valenciano, J., Capobianco, M., and Uribe, J. (2016). Labor vulnerability of Latin American rural women. *Noesis* 26, 130–151. doi: 10.20983/noesis.2017.2.6

Velasco, O. (2016). Juventud y migración. La construcción de una categoría para los derechos humanos. *Inter. Disciplina.* 11, 105–126. doi: 10.22201/ceiich.2448570 5e.2023.29.84482

Villadangos, P. (2022). A defense of administrative rule-making powers. *Rev. Derecho Estado*. 51, 197–226. doi: 10.18601/01229893.n51.07

Villeda, B. (2020). Dinámicas de migración de niños, niñas y adolescentes en Guatemala. Guatemala: Universidad de San Carlos de Guatemala Available at: https://ls3.usac.edu.gt/trabajosocial/iiets/Pub202005.pdf.

Weller, J. (2016). Brechas y transformaciones. La evolución del empleo agropecuario en América Latina. Santiago: CEPAL Available at: https://repositorio.cepal.org/server/api/core/bitstreams/004f4733-552a-43fe-ad4c-1ff3c21dacb3/content.

Zabala, A. (2021). A literature review of rural youth studies in Latin America and the Caribbean during 2001-2019. *Cien. Tecnol. Agropec.* 22, 1–25. doi: 10.21930/rcta.vol22_num3_art:2310